Author’s response to reviews


Authors:

Anette Wennström (anette.wennstrom@odontologi.gu.se)
Margareta Ahlqwist (margareta.ahlqwist@gu.se)
Ulrika Stenman (ulrika.stenman@odontologi.gu.se)
Cecilia Björkelund (cecilia.bjorkelund@allmed.gu.se)
Magnus Hakeberg (magnus.hakeberg@odontologi.gu.se)

Version: 4  Date: 22 October 2013

Author’s response to reviews: see over
Dear Dr Charles Rwenyonyi,

Thank you for accepting our submitted manuscript MS: 1704200390982502. We have now also performed the language corrections requested. We have carefully changed the sentences in the manuscript, as suggested. Please let us know if there are any more corrections that need to be done. In this letter we give responses to these suggestions and explain the changes made in the revised manuscript; “Trends in tooth loss in relation to socio-economic status among Swedish women, aged 38 and 50 years. Repeated cross-sectional surveys 1968-2004.” We hope that you will find these changes in accordance with your intentions.

Language corrections:

1. “Total amount of women” is now changed to “Total number of women…”
   Please see page 6, second paragraph, line 2.

2. “Socio-economy” is now changed to “socio-economic status”
   Please see the abstract on page 2, under the subheading “Background”, last sentence.
   Also on page 5, second paragraph, last sentence.
   And on page 6, under the heading “Non-participation analysis”, first sentence.

3. “Oral status” is now changed to “oral health status”
   Please see the following pages:
   On page 2, subheading “Background”, the second and third sentence.
   On page 4, last paragraph, line 5, 10, 11 and 12.
On page 5, second paragraph, first sentence.

On page 12, under “Discussion”, line 11.

On page 13, first paragraph, last sentence.

Also on page 13, third paragraph, the first and second sentence.

Sincerely

Anette Wennström          Magnus Hakeberg
DDS, doctoral student    Professor, DDS, PhD

Margareta Ahlqwist         Ulrika Stenman        Cecilia Björkelund
Assoc professor, DDS, PhD  DDS                           Professor, MD, PhD

Institute of Odontology, The Sahlgrenska Academy,

University of Gothenburg